Author’s response to reviews

Title: Femoral vein obturator bypass revascularization in groin infectious bleeding: two case reports and review of the literature

Authors:

Albert Busch (busch_a2@chirurgie.uni-wuerzburg.de)
Udo Lorenz (lorenz_u1@chirurgie.uni-wuerzburg.de)
Christian Tiurbe (tiurbe_c@chirurgie.uni-wuerzburg.de)
Christoph Bühler (buehler_c@chirurgie.uni-wuerzburg.de)
Richard Kellersmann (kellersman_r@chirurgie.uni-wuerzburg.de)

Version: 4 Date: 5 December 2012

Author’s response to reviews: see over
Würzburg, December 5th 2012

Journal of Medical Case Reports - Editorial

Cover letter

Subject: MS: 4473367727553837
Femoral vein obturator bypass revascularization in groin infectious bleeding: report of two cases

Dear Reviewers,

Thank you very much for considering our manuscript for publication in the Journal of Medical Case Reports and for your kind suggestions on it.

In addition to submitting a newly revised version of our paper, I’d like to comment on the editors’ concerns.

- In the revised version I rearranged the case presentations section in the abstract in the same way it is organized in the main document.

- I changed the title suffix to “two case reports and review of the literature”, to fit the journal style corresponding to the titles presented in the current issue.

I hope our responses to the editorial team’s comments are clear and sufficient. In case of any further concerns on the manuscript I’m available for correction at any time.

We hope that our manuscript can be further considered for publication.

Sincerely,

Albert Busch